Author's response to reviews

Title: Novel TBX1 loss-of-function mutation causes isolated conotruncal heart defects in Chinese patients without 22q11.2 deletion

Authors:

Yue-Juan Xu (xuyj1006@126.com)
Sun Chen (chengsun@hotmail.com)
Jian Zhang (jzhang.dr@gmail.com)
Shao-Hai Fang (fangshaohai1988@163.com)
Qian-Qian Guo (guoqiangqian1988@sina.com)
Jian Wang (labwangjian@126.com)
Qi-Hua Fu (qihuafu@hotmail.com)
Fen Li (lifen_88@yahoo.com.cn)
Rang Xu (rang_xu@hotmail.com)
Kun Sun (sunkun@xinhuamed.com.cn)

Version: 5 Date: 20 June 2014

Author's response to reviews: see over
Editor of
BMC Medical Genetics

Dear Editor,

We appreciated all the enlightening comments from the editors.
The manuscript has been revised according to the comments. The modified parts were highlighted in blue.
We also responded point by point to editors’ comments as listed below.
Thank you very much.

Best regards

Yours sincerely,

Corresponding author: Kun Sun, Ph.D, Professor
Department of cardiology, Xinhua Hospital
Shanghai Jiaotong University School of Medicine,
1665 KongJiang Road,
Shanghai 200092,
P.R China

Response to editors' comments:

Comments to the Author
1. On p.6, change Ethnics to Ethics.
   Author’s response:
   We have made change in the revised manuscript. Please see the page 6.

2. On p.15, change “We hypothesis” to “We cannot exclude”.
   Author’s response:
   We have made changes in the revised manuscript. Please see the page 15.

3. On p.16, change “Several weaknesses are” to “We recognize some weaknesses that are”.
   Author’s response:
   We have made changes in the revised manuscript. Please see the page 16. Thank you very much.